Author's response to reviews

Title: Personalized bracing plus exercises is effective for adolescent idiopathic scoliosis: prospective cohort study according to the srs and sosort criteria with efficacy and intent-to-treat analysis Negrini S (1,2), Donzelli S (3), Lusini M (3), Minnella S (3), Zaina F (3) (1) Director, Physical and Rehabilitation Medicine, University of Brescia, Italy; (2) IRCCS Don Gnocchi Foundation, Italy; (3) ISICO (Italian Scientific Spine Institute), Milan, Italy

Authors: 
Stefano S Negrini (stefano.negrini@isico.it)
Sabrina S Donzelli (sabrina.donzelli@isico.it)
Monia M Lusini (monia.lusini@isico.it)
Salvatore S Minnella (salvatore.minnella@isico.it)
Fabio F Zaina (fabio.zaina@isico.it)

Version: 2 Date: 6 June 2013

Author’s response to reviews: see over
Dear Chief-Editor

we submit our paper for your evaluation. This is the first prospective study on rigid bracing respecting the SRS and SOSORT criteria for bracing studies, i.e. the criteria proposed by the two main Scientific Societies in this field.

Best regards